Author's response to reviews

Title: Normative data and discriminative properties of short form 36 (SF-36) in Turkish urban population

Authors:

Yucel Demiral (yucel.demiral@deu.edu.tr)
Gul Ergor (gul.ergor@deu.edu.tr)
Belgin Unal (belgin.unal@deu.edu.tr)
Semih Semin (semih.semin@deu.edu.tr)
Yildiz Akvardar (yildiz.akvardar@deu.edu.tr)
Berna Kivircik (berna.kivircik@deu.edu.tr)
Koksal Alptekin (koksal.alptekin@deu.edu.tr)

Version: 6 Date: 16 September 2006

Dear Editor,
 Thank you very much for the information on acceptance of our manuscript (MS:1005918185992970).

We have revised the formatting of the manuscript according to journal checklist. We have also made the following changes according to your suggestions:

1. We have used sentence case in the title and all subheadings
2. "Conclusion" has been changed as "Conclusions".
3. We have removed all highlightings from the MS.
4. We have added the following sentence in Methods section page 4, line 22-24.
   "In order to obtain the data, SF-36 version 1.0 and sociodemographic information questionnaires were completed using a face-to-face interview technique after asking verbal consent from the participants.

5. We have revised The references accordingly and removed the term "et al."

Best wishes

Belgin Unal